Author’s response to reviews

Title: A Rare Symptomatic Congenital Morgagni hernia presenting as a chest pain.

Authors:
Mujtaba Mohamed (mujtaba.mohamed@hackensackmeridian.org)
Alsadiq AL Hillan (alsadiq.alhillan@hackensackmeridian.org)
Jay Shah (jay.shah@hackensackmeridian.org)
Eugene Zurkovsky (doceugenez@gmail.com)
Arif Asif (arif.asif@hackensackmeridian.org)
Mohammad Hossain (mohammad.hossain@hackensackmeridian.org)

Version: 1 Date: 11 Nov 2019

Author’s response to reviews:

Reviewer’s comment: “Add a paragraph at the end of the Introduction that explains why this case report is presented (what is unique and adds to the medical knowledge)

Response: Thanks for comment. As per suggestion, the introduction has been adjusted to clarify the purpose and clinical significance of reporting this case.

Reviewer’s comment: “The case report should include past medical, social, environmental, family and employment history.

Response: Thanks for your comment. We have added the social history, employment and family history to the case report.

Reviewer’s comment: “What medications was the patient receiving prior to diagnosis? Did the patient smoke, and/or consume alcohol?”

Response: Thank you for constructive feedback. We added patient smoking history, Medications he was taking in order to alleviate his pain.

Reviewer’s comment: “Give detailed neurological examination on admission.”

Response: Thanks for this feedback. We added to our case report detailed neurological exams.
Reviewer’s comment:” Give the doses of all medications that were given and their doses”
Response: we mentioned the medication patient was taking to relive his pain.

Reviewer’s comment:” Give all results of laboratory findings (i.e. CBC, liver and renal functions), urinalysis, serology, microbiology etc)”
Response: We mentioned all the laboratory date including CMP, Liver functions, and CBC.

Reviewer’s comment: “Give information about follow-up for at least 6 months. State the time of the follow up”
Response: 6 months follow up was added to the case report.

Reviewer’s comment: “Discussion – add a paragraph at the beginning of the Discussion that summarizes the case and describes what is unique in this case compared to what is available in the literature.”
Response: We appreciate the constructive feedback. We added a two lines at the beginning of discussion as why we are reporting this case.

ALL CHANGES ARE MADE IN A RED COLOR SO IT CAN BE EASILY TRACKED.